Author's response to reviews

Title: Congenital bipartite lunate presenting as a misdiagnosed lunate fracture: a case report

Authors:

Brian WZ Loh (brianwzloh@me.com)
Jason Harvey (harveyj@theparkclinic.com.au)
Eugene TH Ek (eugene_ek@me.com)

Version: 3 Date: 4 November 2010

Author's response to reviews:

Dear Sir/Madam,

I have made the following changes as per the editors' instructions:

Author list correction

Abstract in the manuscript matches exactly the Abstract in the submission system and does not cite references, nor refer to figures or tables

References

Email addresses

Structure: Ensured the Competing interests section precedes the Consent section

Figures: cropped

Additional files: Deleted the additional files section from the manuscript

Track changes/coloured text: Removed all tracking or coloured text from the manuscript

Thank you

Kindest Regards,

Brian Loh